



Renton, W. D., Leveret, H., Guly, C., Smee, H., Leveret, J., & Ramanan, A. V. (2019). Same but different? A thematic analysis on adalimumab biosimilar switching among patients with juvenile idiopathic arthritis. *Pediatric Rheumatology*, *17*, [67 (2019)]. https://doi.org/10.1186/s12969-019-0366-x

Publisher's PDF, also known as Version of record

License (if available): CC BY

Link to published version (if available): 10.1186/s12969-019-0366-x

Link to publication record in Explore Bristol Research PDF-document

This is the final published version of the article (version of record). It first appeared online via BMC at https://pedrheum.biomedcentral.com/articles/10.1186/s12969-019-0366-x#Abs1 . Please refer to any applicable terms of use of the publisher.

University of Bristol - Explore Bristol Research General rights

This document is made available in accordance with publisher policies. Please cite only the published version using the reference above. Full terms of use are available: http://www.bristol.ac.uk/pure/about/ebr-terms

RESEARCH ARTICLE

Same but different? A thematic analysis on adalimumab biosimilar switching among patients with juvenile idiopathic arthritis

William D. Renton¹, Helen Leveret¹, Catherine Guly², Heather Smee¹, Jamie Leveret¹ and Athimalaipet V. Ramanan^{1,3*}

Abstract

Background: Biologic medications have dramatically enhanced the treatment of many chronic paediatric inflammatory conditions. Their high cost is a factor that prohibits their broader use. Cheaper generic versions, or biosimilars, are increasingly being used. Healthcare services are switching some patients over to biosimilars for economic reasons, known as 'non-medical switching'. Some patients unsuccessfully switch due to perceived decreases in efficacy or non-specific drug effects. The implications of failed switching include exhaustion of therapeutic options, unnecessary exposure to other medications, increased healthcare utilisation, worse patient outcomes and higher overall healthcare costs. Patient perceptions almost certainly play a role in these 'failed switches'.

Methods: A thematic analysis was performed to better understand patient and parent perceptions on non-medical biosimilar switching. The study was conducted in accordance with the Consolidated Criteria for Reporting Qualitative Research recommendations. Patients with juvenile idiopathic arthritis currently taking adalimumab were included.

Results: Nine families were interviewed just prior to a hospital trust-wide non-medical switch to an adalimumab biosimilar. Several common themes were identified. The most frequent concerns were regarding practical aspects of the switch including the medication administration device type; the colour of the medication and administration device; and whether the injections would sting more. The relative safety and efficacy of the biosimilar was raised although most families felt that there would be no significant difference. Anxieties about the switch were largely placated by reassurances from the medical team.

Conclusions: We derived recommendations based on existing adult literature and the observations from our study to optimise the benefits from non-medical biosimilar switching.

Keywords: Paediatric rheumatology, Juvenile idiopathic arthritis, Uveitis, Biosimilars, Adalimumab, Qualitative

Background

Biologic medications, including monoclonal antibodies, are medications derived from living organisms. These medications, including adalimumab, have dramatically improved outcomes of chronic inflammatory conditions

* Correspondence: avramanan@hotmail.com

Full list of author information is available at the end of the article

including refractory juvenile idiopathic arthritis (JIA) [1, 2] and JIA associated uveitis [3, 4].

Biologics are expensive and their cost is a factor that prohibits their broader use. Many index biologics (biooriginators) are still subject to copyright patents, contributing to their high cost. However, for several biologics, generic versions (biosimilars) are becoming available. Unlike conventional medications, biosimilars are not considered completely equivalent to their biooriginator as they are large and complex molecules that

© The Author(s). 2019 **Open Access** This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated.







¹Bristol Royal Hospital for Children, University Hospitals Bristol NHS Foundation Trust, Bristol, UK

 $^{^3\}mathrm{Translational}$ Health Sciences, Bristol Medical School, University of Bristol, Bristol, UK

are very sensitive to any slight change in the manufacturing process [5].

Biosimilar developers must demonstrate that their biosimilar is highly similar to the bio-originator (notwithstanding normal variability inherent to all biologics) and that there are no clinically meaningful differences regarding quality, safety and efficacy [6, 7]. Regulating bodies, including the European Medicines Agency, and rheumatology groups have encouraged a Bayesian approach to the development of biosimilars in order to abbreviate licencing pathways, help lower costs and increase access to these medications [8–10]. Data for one indication may be extrapolated to others (assuming the same mechanism of action is used), again easing the statistical threshold and abbreviating the approval process [11, 12].

Theoretical concerns when switching to biosimilars include a loss of efficacy, changes in immunogenicity (including the development of anti-drug antibodies) and differences in the safety profile compared with the biooriginator [13]. Despite these apprehensions, outcomes from blinded, randomized, controlled trials in adults have been reassuring [14]. While this is the case, large scale paediatric trials are lacking. Nonetheless, healthcare services are tending towards switching patients to biosimilars for economic reasons, known as 'non-medical switching' [13].

Experience among adults suggests that the uptake of biosimilars in open label environments is hindered when compared to blinded trials. These 'failed switches' are usually attributed to subjective reports of perceived decrease in efficacy or non-specific drug effects [15–17]. These are thought to largely be due to the nocebo effect; noxious reactions to therapeutic interventions that occur because of negative expectations of the patient [18]. Emerging paediatric data, while scarce, suggests that some children also unsuccessfully switch [19]. The implications of failed switching could potentially include exhaustion of therapeutic options, unnecessary exposure to other medications, increased healthcare utilisation, worse patient outcomes and higher overall healthcare costs.

It is hypothesised that patient perceptions strongly influence failed biosimilar switching [20].

Methods

This study aims to develop an understanding of the perceptions of paediatric patients and their parents with regard to biosimilar switching. A thematic analysis was performed. Patients with a diagnosis of JIA, under the age of 18 years, on adalimumab (a fortnightly subcutaneous injection) were included. All families were English speaking and literate. They were recruited from paediatric rheumatology outpatient clinics at the Bristol Children's Hospital and Bristol Eye Hospital, tertiary hospitals in the United Kingdom, over a two-week period in December 2018. The study was performed prior to a trust-wide mandatory change from the adalimumab bio-originator to a biosimilar on guidance from the National Health Service (NHS).

Patients were invited to participate either by telephone (5 families) or at the time of clinical appointments (4 families). Convenience sampling was predominantly used with supplementary purposive sampling (aiming to include a range of patients representative of the typical distribution of age, gender and disease severity). Interviews were conducted in a private setting at the outpatient department on the same day as scheduled appointments if possible.

Two researchers with experience in qualitative research methods, but without expertise in paediatric rheumatology or biologic medications (HL and JL) ran the patient led, semi-structured interviews. They had not previously met the families. Participants understood the purpose of the study and that the interviewers were not part of the treating team, nor experts in paediatric rheumatology or biologics. Individual families were interviewed on a single occasion and had no further active participation in the study following the interviews. Patients were interviewed together with their parent(s). Questions were directed to all participants and specifically to the patient if they did not volunteer an answer. Prompting questions were used if required (Additional file 1). Families were provided a copy of a plain language summary outlining the process of transition to the biosimilar (designed by the trust pharmacy and paediatric rheumatology team) on the day of the interviews (Additional file 2).

Interviews were audiotaped and transcribed verbatim. Field notes were also taken during the interviews. Names and identifying features were removed from the transcripts for anonymity.

An inductive approach was taken; identifying and developing themes from the data [21, 22]. Coding and analysis were conducted by HL and JL using a qualitative analysis computer program (NVivo 12.2.0, QSR International). WR separately coded and analysed the data using the same program. Analyses were compared and integrated to create unifying themes.

Informed, written consent was obtained at the time of recruitment. The study was prospectively approved by the University Hospitals Bristol Questionnaire, Interview & Survey Group. The study was conducted and written in accordance with the Consolidated Criteria for Reporting Qualitative Research (COREQ) recommendations [23].

Results

Of 10 families invited to participate, 9 consented and completed the interviews (1 declined due to time constraints). Eight interviews were conducted with the child and at least 1 parent contributing, while 1 interview was conducted with a patient's parents only. All patients had a diagnosis of JIA with associated uveitis. Patient demographics are detailed in Table 1.

Interviews lasted between 6 and 51 min (mean 17). All patients and parents involved made significant contributions to the interview answers. Most families raised some concerns but had an overall positive or apathetic view on the switch. One family was strongly dissatisfied about the non-optional change.

The study identified five main themes concerning patients' and families' perceptions of the switch to a biosimilar medication; 'drug administration, 'concerns', 'benefits', 'equivalence', and 'trust in treating clinicians'. Some themes contained clear associated sub-themes outlined in the thematic schema (Fig. 1).

Drug administration

Issues surrounding practicalities of drug administration were important to most respondents. These issues resonated with the patients more than other issues which their parents were more concerned about.

Patients were anxious about having access to the same medication administration device (the bio-originator is available as either a prefilled syringe or prefilled auto-injecting pen) and the potential of having to use an unfamiliar device. Past versions of the adalimumab bio-originator contained a citrate preservative which contributed to stinging as the medication is injected. Several patients recall what this felt like and those that have never been on a stingy preparation were aware of its existence. Whether the new medication would sting was a prime concern, although some patients were able to balance a possible worse outcome on an individual level against a perceived benefit for society.

that." (parent)

"Your concerns as well have been about 'just tell me it's not going to hurt anymore' or ... about whether it's stingy or not." (parent to patient)

"How would you feel about taking it if it did hurt? (interviewer) "I think it probably be good for society so I guess I would probably do it." (patient)

The colour of a biosimilar medication, administration device and its packaging was a major concern for most families. Several families specified concerns about the colour yellow (no other colours were specified), most attributing this to their previous experience with methotrexate.

"... if it is yellow, he's not going to go near it." (parent)

 Table 1 Patient demographic details

Age	Gender	Time since diagnosis (months)	Time since commencing adalimumab (months)	Ocular complications	Active joint count at time of review	Uveitis activity at time of review *
6	F	37	24	Nil	0	Right 0 Left 0
8	F	72	13	Nil	0	Right 2+ Left 2+
10	Μ	66	58	Cataract, posterior synechiae, ocular hypertension	0	Right 0.5+ Left 0.5+
11	F	82	27	Nil	1	Right 0.5+ Left 0.5+
12	Μ	43	25	Cataract	0	Right 0.5+ Left 0.5+
12	F	127	80	Posterior synechiae	0	Right 0 Left 0
13	Μ	143	122	Cataract, posterior synechiae, band keratopathy	0	Right 0 Left 0.5+
15	F	172	21	Nil	0	Right 0.5+ Left 0.5+
17	F	190	59	Nil	0	Right 0.5+

*anterior chamber cell grading as per standardization of uveitis nomenclature (SUN) criteria [24]



" [if] it is yellow and he can see it is yellow then that will make him gag because it is a psychological thing." (parent)

Concerns

Within the theme of 'concerns', anxiety about side effects was frequently expressed. There was anxiety about switching from the known to the unknown. For other families, provided there were no new side effects, they didn't have any significant concerns about switching to a biosimilar medication.

"It's not ideal because you know that something works and it really is just to know that it's not going to have the side effects" (parent)

"We understand it, as long as he doesn't have any side effects detrimental to him." (parent)

Several families were concerned about what would happen if there were problems with the switch and whether it would be possible to switch back in the event of a poor outcome.

"If we swap and it is worse, and it doesn't do anything, what happens then? Are we then stuck effectively with ... the cheaper product?" (parent)

Some were concerned about the logistics of switching to a biosimilar and whether the home delivery process would be disrupted, especially given the need to store these medications in a refrigerator.

"My main concern is that the smooth transition goes through to [delivery service provider] because they are not always accurate with the delivery time." (parent)

There was some uncertainty about whether a biosimilar would be as effective as the bio-originator. Several patients referred to the fact that as it is cheaper, it may be inferior. However, many acknowledged that although this might be a perception, when they explored this thinking they could rationalise that just because a product is cheaper it does not equate to inferiority.

"A lot of people say if it is cheaper then it is not as good as the most expensive one." (parent)

"Just because it is cheaper doesn't mean that it is inferior." (parent)

Two families suggested switching adult patients prior to paediatric patients to help mitigate some of these concerns.

"If you want to change, change the adults but don't change the children" (parent)

Benefits

Almost all respondents identified cost savings as a benefit of switching to a biosimilar (of note, this was mentioned in the pharmacy letter and prompting questions).

"I think it is actually probably really good because I do not want the NHS to like not work so I think it is really good that we are going to a cheaper version." (patient)

Others identified benefits including greater accessibility to the drug. Increased access felt important for some who remember the challenges of getting biological medication for their child and the impact that doing so had on their lives. Some families highlighted that cost savings could mean that resources could be used elsewhere.

"Now that there is somebody else out there they might not have to fight to get the drug that their kids deserve. It is good." (parent) Many patients were theoretically willing to accept the risk of possible side-effects for the good of other patients. Patients felt that research data would be another by-product of the switch and were generally positive about contributing towards this.

"The competition, you have got more opportunities for people to ... explore and push the boundaries a bit more in terms of research." (parent)

Equivalence

Although there were anxieties about a biosimilar medication being inferior, a significant majority of respondents expressed that they thought the biosimilar medication would probably be similarly safe and effective.

"For me, it is the same drug but by a different supplier so it doesn't really make any difference so I do not mind at all." (parent)

Even respondents who had significant anxieties about inferiority also simultaneously acknowledged there may be no difference in efficacy. This uncertainty was a relatively universal finding.

"... as long as it works as well and I guess until we have tried it, we won't feel reassured about that" (parent)

Trust in clinical staff

For many respondents, their anxieties and concerns were mitigated by their faith in their medical team.

"They wouldn't let you have something that's not going to work." (parent)

Even families who were very unhappy about the switch did not blame the medical team and felt that the decision was out of the hands of clinical staff. Frustration regarding the forced switch was an uncommon finding and was generally directed towards the pharmacy, the hospital trust administrators and the NHS.

Most patients had informally heard about the upcoming switch through their treating team, disease support groups or social media. Feedback on the hospital trust pharmacy letter was generally very positive although several patients suggested that a phone call or face to face notification by a familiar member of the treating team may be more effective in the first instance.

Discussion

To our knowledge this is the first study exploring perceptions on switching to biosimilar medications among paediatric patients and their parents. Achieving data saturation was not a prospective target although clear thematic patterns emerged. While there were minor differences in the specific coding nomenclature and category structure, the two separate analyses identified very similar themes despite separate areas of expertise among the analysts.

The researchers anticipated that the major concerns would be regarding the safety and efficacy of these new mediations. While these issues arose, the overall sentiment was that things would probably be fine and most families were understanding. Patients' uncertainties appeared to be placated by their trust in the treating team. Altruism was a prominent finding with most families happy to accept the uncertainties of switching for the greater good of the healthcare system.

The much more prevalent concerns had little to do with the actual medication. Patients in particular were most worried about the type of medication administration device, the presence or absence of a stinging preservative and the colour of the medication and packaging. Views were strongly shaped by previous experiences; this was particularly apparent with concerns about the colour of the new product. By the nature of these patients being on adalimumab they will all have used methotrexate (a yellow liquid often with prominent yellow branding and administration devices) in the past or as part of their current treatment regimen. Many families who raised this concern explicitly referred to their previous experience with methotrexate, its associated side-effects and the psychological sequelae from this.

'Failed switches' are not uncommon among adults undergoing non-medical monoclonal antibody biosimilar switching with the nocebo factor playing a significant role [20]. Previous experience and expectation significantly affect noxious symptoms attributable to the nocebo affect [25]. Effective communication strategies have been identified as vital in minimizing this phenomenon among adult patients [26, 27]. The themes identified in our study may assist with proactively addressing patient expectations during medication counselling and patient education programs in order to minimise the incidence of failed switching and contribute to an improved patient experience.

It's known that clinicians often disagree with nonmedical switching and feel that patients should be given a choice [28]. Nonetheless, it is important to portray a positive message despite any clinician opinions and biases. We have outlined recommendations on the basis of existing adult literature and the observations from this study (Table 2).

This study is not without limitations. Participants' responses would be influenced by multiple local

Table 2 Recommendations for non-medical paediatric biosimilar switching

Recommendations for non-medical paediatric biosimilar switching

- Adequate biosimilar teaching for all staff
- A coherent message from all members of the team (medical, nursing, pharmacy, others)
- Written and verbal patient education prior to switching
- Disease and medication specific information to be provided to families
- Honest and balanced yet positive framing of information
- Detailed information on practical aspects of the biosimilar including
- device types, additives (including preservatives) and delivery logistics
- Information to be written and endorsed by familiar treating team members rather than anonymous administrators/pharmacy

factors, through both informal (discussions with the treating team, local media, social media, support groups) and formal (pharmacy letter) means. Participant bias may have contributed to the positive outlook. Purposive selection is subject to selection bias although this was partially offset by the use of supplementary random sampling for the majority of patients. The data is also limited to a specific medication used among a single disease at one centre. Despite these factors we believe that the findings are generalisable to other healthcare settings and other chronic medical conditions treated with biologics and therapeutic proteins.

Conclusions

With significant financial strain across many healthcare settings, non-medical biosimilar switching is likely to be an increasingly frequent occurrence. We hope that the findings and recommendations from this study may help clinicians assist their patients through this process.

Additional files

Additional file 1: Prompting questions. (DOCX 23 kb) Additional file 2: Pharmacy letter. (DOCX 23 kb)

Abbreviations

COREQ: Consolidated Criteria for Reporting Qualitative Research; JIA: Juvenile Idiopathic Arthritis; NHS: National Health Service; SUN: Standardization of Uveitis Nomenclature

Acknowledgements

The authors would like to thank all of the patients and their families for their generous contributions to this work.

Author's contributions

WDR and AVR initiated the study, performed a literature review, designed the study, contributed to writing the paper and final manuscript editing. HL and JL contributed to the study design, conducted patient interviews, contributed to writing the paper and final manuscript editing. HL, JL and WDR coded the transcripts and performed the thematic analysis. HS and CG assisted with patient recruitment and final manuscript editing. All authors read and approved the final manuscript.

Funding

No funding was received for this study.

Availability of data and materials

The datasets generated and/or analysed during the current study are not publicly available to protect individual privacy but are available from the corresponding author on reasonable request.

Ethics approval and consent to participate

The study was prospectively approved by the University Hospitals Bristol Questionnaire, Interview & Survey Group. Informed, written consent to participate was obtained at the time of recruitment.

Consent for publication

Informed, written consent to publish anonymised data was obtained at the time of recruitment.

Competing interests

No competing interests are declared.

Author details

¹Bristol Royal Hospital for Children, University Hospitals Bristol NHS Foundation Trust, Bristol, UK. ²Bristol Eye Hospital, University Hospitals Bristol NHS Foundation Trust, Bristol, UK. ³Translational Health Sciences, Bristol Medical School, University of Bristol, Bristol, UK.

Received: 3 May 2019 Accepted: 5 September 2019 Published online: 04 October 2019

References

- Lovell DJ, Ruperto N, Goodman S, Reiff A, Jung L, Jarosova K, et al. Adalimumab with or without methotrexate in juvenile rheumatoid arthritis. N Engl J Med. 2008;359(8):810–20.
- Guzman J, Oen K, Tucker LB, Huber AM, Shiff N, Boire G, et al. The outcomes of juvenile idiopathic arthritis in children managed with contemporary treatments: results from the ReACCh-out cohort. Ann Rheum Dis. 2015;74(10):1854–60.
- Ramanan AV, Dick AD, Jones AP, McKay A, Williamson PR, Compeyrot-Lacassagne S, et al. Adalimumab plus methotrexate for uveitis in juvenile idiopathic arthritis. N Engl J Med. 2017;376(17):1637–46.
- Cann M, Ramanan AV, Crawford A, Dick AD, Clarke SLN, Rashed F, et al. Outcomes of non-infectious Paediatric uveitis in the era of biologic therapy. Pediatr Rheumatol Online J. 2018;16(1):51.
- Vulto AG, Jaquez OA. The process defines the product: what really matters in biosimilar design and production? Rheumatology (Oxford). 2017; 56(suppl_4):iv14–29.
- European Medicines Agency (EMA). Biosimilar medicines: overview 2018 [cited 2018 12/12/2018]. Available from: https://www.ema.europa.eu/en/ human-regulatory/overview/biosimilar-medicines.
- Goel N, Chance K. Biosimilars in rheumatology: understanding the rigor of their development. Rheumatology (Oxford). 2017;56(2):187–97.
- Weiss RE, Xia X, Zhang N, Wang H, Chi E. Bayesian methods for analysis of biosimilar phase III trials. Stat Med. 2018;37(20):2938–53.
- Ruperto N, Brunner HI, Lovell DJ, Martini A. Paediatric rheumatology international trials O, the pediatric rheumatology collaborative study G. extrapolation or controlled trials in paediatrics: the current dilemma. Arch Dis Child. 2017;102(10):949–51.
- Kay J, Schoels MM, Dorner T, Emery P, Kvien TK, Smolen JS, et al. Consensus-based recommendations for the use of biosimilars to treat rheumatological diseases. Ann Rheum Dis. 2018;77(2):165–74.
- 11. Lee H. Is extrapolation of the safety and efficacy data in one indication to another appropriate for biosimilars? AAPS J. 2014;16(1):22–6.
- Bridges SL Jr, White DW, Worthing AB, Gravallese EM, O'Dell JR, Nola K, et al. The science behind Biosimilars: entering a new era of biologic therapy. Arthritis Rheumatol. 2018;70(3):334–44.
- Azevedo V, Dörner T, Strohal R, Isaacs J, Castañeda-Hernández G, Gonçalves J, et al. Biosimilars: considerations for clinical practice. Considerations in Medicine. 2017;1(1):13–8.
- Komaki Y, Yamada A, Komaki F, Kudaravalli P, Micic D, Ido A, et al. Efficacy, safety and pharmacokinetics of biosimilars of anti-tumor necrosis factoralpha agents in rheumatic diseases; a systematic review and meta-analysis. J Autoimmun. 2017;79:4–16.
- 15. Glintborg B, Loft AG, Omerovic E, Hendricks O, Linauskas A, Espesen J, et al. To switch or not to switch: results of a nationwide guideline of mandatory

switching from originator to biosimilar etanercept. One-year treatment outcomes in 2061 patients with inflammatory arthritis from the DANBIO registry. Ann Rheum Dis. 2018.

- Avouac J, Molto A, Abitbol V, Etcheto A, Salcion A, Gutermann L, et al. Systematic switch from innovator infliximab to biosimilar infliximab in inflammatory chronic diseases in daily clinical practice: the experience of Cochin University hospital, Paris. France Semin Arthritis Rheum. 2018;47(5):741–8.
- Tweehuysen L, van den Bemt BJF, van Ingen IL, de Jong AJL, van der Laan WH, van den Hoogen FHJ, et al. Subjective complaints as the Main reason for biosimilar discontinuation after open-label transition from reference infliximab to biosimilar infliximab. Arthritis Rheumatol. 2018;70(1):60–8.
- Benedetti F, Lanotte M, Lopiano L, Colloca L. When words are painful: unraveling the mechanisms of the nocebo effect. Neuroscience. 2007;147(2): 260–71.
- Decock D, Davies R, Kearsley-Fleet L, Baildam E, Beresford M, Foster H, et al. P34 Biosimilar use in children and young people with juvenile idiopathic arthritis in a real-world setting in the United Kingdom. Rheumatology. 2018; 57(suppl_8):key273.036-key273.036.
- Kravvariti E, Kitas GD, Mitsikostas DD, Sfikakis PP. Nocebos in rheumatology: emerging concepts and their implications for clinical practice. Nat Rev Rheumatol. 2018.
- 21. Braun V, Clarke V. Successful qualitative research: a practical guide for beginners: sage; 2013.
- 22. Crabtree BF, Miller WL. Doing qualitative research: Sage publications; 1999.
- Tong A, Sainsbury P, Craig J. Consolidated criteria for reporting qualitative research (COREQ): a 32-item checklist for interviews and focus groups. Int J Qual Health Care. 2007;19(6):349–57.
- Jabs DA, Nussenblatt RB, Rosenbaum JT. Standardization of uveitis nomenclature working G. standardization of uveitis nomenclature for reporting clinical data. Results of the first international workshop. Am J Ophthalmol. 2005;140(3):509–16.
- Reicherts P, Gerdes AB, Pauli P, Wieser MJ. Psychological placebo and nocebo effects on pain rely on expectation and previous experience. J Pain. 2016;17(2):203–14.
- 26. Jørgensen TS, Skougaard M, Asmussen HC, Lee A, Taylor PC, Gudbergsen H, et al. Communication strategies are highly important to avoid nocebo effect when performing non-medical switch from originator product to biosimilar product: Danish results from applying the Parker model a qualitative 3-step research model. Arthritis Rheumatol. 2017;69.
- Kristensen LE, Alten R, Puig L, Philipp S, Kvien TK, Mangues MA, et al. Nonpharmacological effects in switching medication: the nocebo effect in switching from originator to biosimilar agent. BioDrugs. 2018;32(5):397–404.
- Aladul MI, Fitzpatrick RW, Chapman SR. Healthcare professionals' perceptions and perspectives on biosimilar medicines and the barriers and facilitators to their prescribing in UK: a qualitative study. BMJ Open. 2018; 8(11):e023603.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

